\documentclass[a4paper,10pt,oneside]{article}

\marginparwidth=15pt
\oddsidemargin=10pt
\textwidth=440pt
\textheight=600pt

% PACKAGES
\usepackage[latin1]{inputenc}


% RAVI's FONT.
%\usepackage{mathpazo}
%\usepackage{euler}


\newcommand{\marginalfootnote}[1]{
   \footnote{#1}
   \marginpar{\hfill {\sf\thefootnote}}}
\newcommand{\edit}[1]{\marginalfootnote{#1}}


\usepackage[T1]{fontenc}
\usepackage[UKenglish]{babel}
\usepackage{amsfonts}
\usepackage{amssymb}
\usepackage{amsthm}
\usepackage{graphicx}
\usepackage{fancyhdr}
\usepackage{color}
\usepackage{makeidx}
\usepackage{amsmath}
\usepackage[all,cmtip]{xy}
\usepackage{blindtext}
\usepackage{parskip}
\usepackage{hyperref}
\usepackage[notlof,notlot]{tocbibind}
\usepackage{mathrsfs}



\usepackage{tikz}
%%%<
\usepackage{verbatim}
\usepackage{verbatim}
%\usepackage[active,tightpage]{preview}
%\PreviewEnvironment{tikzpicture}
%\setlength\PreviewBorder{5pt}%
%%%>




\addtocounter{section}{14}

%\setlength{\parindent}{0pt}
%\setlength{\headheight}{15.2pt}
%\renewcommand{\sectionmark}[1]{\markboth{#1}{}}
%\lhead{{\thepage}}
%\chead{}
%\rhead{{\textup{\leftmark}}}
%\lfoot{}
%\cfoot{}
%\rfoot{}

\makeindex

\definecolor{red}{rgb}{1,0,0}
\definecolor{lblue}{rgb}{0.3,0.4,1}
\definecolor{orange}{rgb}{1,0.6,0.2}
\definecolor{green}{rgb}{0.1,0.7,0.1}

% MEANINGS
\newtheoremstyle{thm}% name of the style to be used
  {3pt}% measure of space to leave above the theorem. E.g.: 3pt
  {3pt}% measure of space to leave below the theorem. E.g.: 3pt
  {\em}% name of font to use in the body of the theorem
  {0pt}% measure of space to indent
  {\bfseries}% name of head font
  {}% punctuation between head and body
  {5pt}% space after theorem head
  {}% Manually specify head
\newtheoremstyle{rem}% name of the style to be used
  {3pt}% measure of space to leave above the theorem. E.g.: 3pt
  {3pt}% measure of space to leave below the theorem. E.g.: 3pt
  {}% name of font to use in the body of the theorem
  {0pt}% measure of space to indent
  {\bfseries}% name of head font
  {}% punctuation between head and body
  {5pt}% space after theorem head
  {}% Manually specify head


% MACROS
\theoremstyle{thm}
%\newtheorem*{them}{Theorem}
\newtheorem{thm}{Theorem}[subsection]
\newtheorem{prop}[thm]{Proposition}%[subsection]
%\newtheorem*{propn}{Proposition}
\newtheorem{cor}[thm]{Corollary}%[subsection]
\newtheorem{lemma}[thm]{Lemma}%[subsection]
\theoremstyle{rem}
\newtheorem{eg}[thm]{Example}%[subsection]
\newtheorem{defin}[thm]{Definition}%[subsection]
\newtheorem{rem}[thm]{Remark}%[subsection]


% ADDITIONAL SYMBOLS
\newcommand\ab{{\rm ab}}
\newcommand\ad{{\rm ad}}
\newcommand\aff{{\rm aff}}
\newcommand\alg{{\rm alg}}
\newcommand\ant{{\rm ant}}
\newcommand\codim{{\rm codim}}
\newcommand\diag{{\rm diag}}
\newcommand\et{{\rm et}}
\newcommand\gr{{\rm gr}}
\newcommand\gp{{\rm gp}}
\newcommand\coker{{\rm coker}}
\newcommand\id{{\rm id}}
\newcommand\lin{{\rm lin}}
\renewcommand\log{{\rm log}}
\newcommand\op{{\rm op}}
\newcommand\ord{{\rm ord}}
\newcommand\red{{\rm red}}
\newcommand\reg{{\rm reg}}
\newcommand\rk{{\rm rk}}

\newcommand\Aff{{\rm Aff}}
\newcommand\Aut{{\rm Aut}}
\newcommand\Bl{{\rm Bl}}
\newcommand\End{{\rm End}}
\newcommand\GL{{\rm GL}}
\newcommand\Ga{{\mathbb G}_a}
\newcommand\Gm{{\mathbb G}_m}
\newcommand\Gal{{\rm Gal}}
\newcommand\Gr{{\rm Gr}}

\newcommand\Hilb{{\rm Hilb}}
\newcommand\Quot{{\rm Quot}}


\newcommand\fX{{\mathfrak{X}}}
\newcommand\fY{{\mathfrak{Y}}}
\newcommand\ccF{{\rm \mathcal{F}}}
\newcommand\ccD{{\rm \mathcal{D}}}
\newcommand\ccO{{\rm \mathcal{O}}}

\newcommand\Spf{\mathrm{Spf}}
\newcommand\cSpec{\underline{\mathrm{Spec}}}
\newcommand\disc{{\mathrm{disc}}}
\newcommand\Frob{{\mathrm{Frob}}}
\newcommand\rR{{\mathrm{R}}}
\newcommand\Tate{{\mathrm{Tate}}}
\newcommand\fG{{\mathfrak{G}}}
\newcommand\fH{{\mathfrak{H}}}

\newcommand\Ann{{\rm Ann}}
\newcommand\Spv{{\rm Spv}}
\newcommand\RZ{{\rm RZ}}
\newcommand\Cont{{\rm Cont}}
\newcommand\Hom{{\rm Hom}}
\newcommand\Ind{{\rm Ind}}
\newcommand\Irr{{\rm Irr}}
\newcommand\Lie{{\rm Lie}}
\newcommand\M{{\rm M}}
\newcommand\Mor{{\rm Mor}}
\newcommand\Pic{{\rm Pic}}
\newcommand\Proj{{\rm Proj}}
\newcommand\PGL{{\rm PGL}}
\newcommand\PSL{{\rm PSL}}
\newcommand\SL{{\rm SL}}
\newcommand\Spec{{\rm Spec}}
\newcommand\Spa{{\rm Spa}}
\newcommand\Sp{{\rm Sp}}
\newcommand\Supp{{\rm Supp}}
\newcommand\Sym{{\rm Sym}}
\newcommand\Univ{{\rm Univ}}
\newcommand\Ass{{\rm Ass}}
\newcommand\Frac{{\rm Frac}}
\newcommand\Tor{{\rm Tor}}
\newcommand\cTor{{\mathcal Tor}}
\newcommand\cHom{{\underline{\rm Hom}}}
\newcommand\RSchemes{{\bfseries aff} $R$-{\bfseries Sch}}
\newcommand\Schemes{{\mathbf{Sch}}}
\newcommand\Top{{\mathbf{Top}}}
\newcommand\Cov{{\mathrm{Cov}}}
\newcommand\Sets{\mathbf{Set}}
\newcommand\Func{\mathbf{Func}}
\newcommand\Pref{{\bfseries Presh}_X}
\newcommand\Sheaf{{\bfseries She}_X)}
\newcommand\Shv{\mathbf{Shv}}
\newcommand\PShv{\mathbf{PShv}}
\newcommand\Open{{\mathbf{Op}}_X}
\newcommand\Ab{{\mathbf{Ab}}}
\newcommand\Grps{\mathbf{Grp}}
\newcommand\RMod{\mathbf{Mod}_R}
\newcommand\RAlg{\mathbf{Alg}_R}
\newcommand\Ob{\textup{Ob}}
\newcommand\bA{{\mathbb A}}
\newcommand\bC{{\mathbf C}}
\newcommand\bF{{\mathbf F}}
\newcommand\bG{{\mathbb G}}
\newcommand\bI{{\mathbf I}}
\newcommand\bN{{\mathbf N}}
\newcommand\bP{{\mathbb P}}
\newcommand\bQ{{\mathbf Q}}
\newcommand\bR{{\mathbf R}}
\newcommand\bS{{\mathbb S}}
\newcommand\bV{{\mathbf V}}
\newcommand\bX{{\mathbf X}}
\newcommand\bZ{{\mathbf Z}}

\newcommand\bmu{{\mathbb \mu}}

\newcommand\cA{{\mathscr A}}
\newcommand\cB{{\mathscr B}}
\newcommand\cC{{\mathscr C}}
\newcommand\cD{{\mathscr D}}
\newcommand\cE{{\mathscr E}}
\newcommand\cF{{\mathscr F}}
\newcommand\cG{{\mathscr G}}
\newcommand\cH{{\mathscr H}}
\newcommand\cI{{\mathscr I}}
\newcommand\cJ{{\mathscr J}}
\newcommand\cL{{\mathscr L}}
\newcommand\cM{{\mathscr M}}
\newcommand\cN{{\mathscr N}}
\newcommand\cO{{\mathscr O}}
\newcommand\cQ{{\mathscr Q}}
\newcommand\cR{{\mathscr R}}
\newcommand\cS{{\mathscr S}}
\newcommand\cT{{\mathscr T}}
\newcommand\cV{{\mathscr V}}
\newcommand\cW{{\mathscr W}}
\newcommand\cX{{\mathscr X}}
\newcommand\cY{{\mathscr Y}}
\newcommand\cZ{{\mathscr Z}}
\newcommand\cProj{{\mathcal Proj}}
\newcommand\fg{{\mathfrak g}}
\newcommand\fh{{\mathfrak h}}
\newcommand\ft{{\mathfrak t}}
\newcommand\fD{{\mathfrak D}}
\newcommand\fS{{\mathfrak S}}
\newcommand\fm{{\mathfrak m}}
\newcommand\fn{{\mathfrak n}}
\newcommand\fp{{\mathfrak p}}
\newcommand\fq{{\mathfrak q}}
\newcommand\ccU{{\mathfrak U}}
\newcommand\cU{{\mathscr U}}
\newcommand\bJ{{\mathbf J}}
\newcommand\bH{{\mathbf H}}
\newcommand\fA{{\mathfrak A}}
\newcommand\fB{{\mathfrak B}}
\newcommand\fC{{\mathfrak C}}
\newcommand\SR{{\fS(R_{\mathrm{fpqc}})}} % descent 1


\newcommand\tg{\tilde{g}}
\renewcommand\th{\tilde{h}}
\newcommand\tlambda{\tilde{\lambda}}
\newcommand\tp{\tilde{p}}
\newcommand\tq{\tilde{q}}

\newcommand\tG{\tilde{G}}
\newcommand\tM{\tilde{M}}
\newcommand\tV{\tilde{V}}
\newcommand\tX{\tilde{X}}
\newcommand\tY{\tilde{Y}}
\newcommand\tZ{\tilde{Z}}

\newcommand\tcZ{\tilde{\mathcal Z}}

\newcommand\ualpha{\underline{\alpha}}
\newcommand\ulambda{\underline{\lambda}}

\newcommand\BT{\mathbf{BT}}
\newcommand\fotimes{{\widehat{\otimes}}}

\newcommand\Inf{{\mathrm{Inf}}}
\newcommand\uomega{{\underline{\omega}}}
\newcommand\Ext{{\mathrm{Ext}}}
\newcommand\rH{{\rm H}}
\newcommand\rV{{\rm V}}
\newcommand\rT{{\rm T}}
\newcommand\fa{{\mathfrak a}}
\newcommand\rad{{\mathrm{rad}}}
\newcommand\bD{{\mathbf{D}}}
\newcommand\ccG{{\mathcal{G}}}



%\renewcommand{\sectionmark}[1]{\markright{\thesection\ #1}}
%\fancyhf{}  % delete current setting for header and footer
%\fancyhead[LE,RO]{\thepage}
%\fancyhead[LO]{\bfseries\rightmark}
%\fancyhead[RE]{\bfseries\leftmark}
%\renewcommand{\headrulewidth}{0.5pt}
%\renewcommand{\footrulewidth}{0pt}
%\addtolength{\headheight}{0.5pt} % make space for the rule
%\fancypagestyle{plain}{%
%   \fancyhead{} % get rid of headers on plain pages
 %  \renewcommand{\headrulewidth}{0pt} % and the line
%}

\begin{document}
\begin{thebibliography}{dieci}
\bibitem[BGR]{BGR} S.\,Bosch, U.\,G\"untzer,\,R.\,Remmert, {\em Non-archimedean analysis}, Springer-Verlag.
%\bibitem[Bosch]{Bosch} S. Bosch, {\em Lectures on Formal and Rigid Geometry}. Springer Lecture Notes in Mathematics, vol. 2105.
%\bibitem[Con]{Con} B. Conrad - ``Several approached to non-archimedean geometry'', Arizona Winter School lecture notes available at: \texttt{http://math.stanford.edu/$\sim$conrad/papers/aws.pdf}.
\bibitem[EGA]{EGA} A. Grothendieck, EGA, Publ.\,Math.\,IHES 1961--68.
\bibitem[SGA4]{SGA4} A. Grothendieck, SGA4.
\bibitem[Mat]{Mat} H. Matsumura - {\em Commutative Ring Theory}. Vol. 8 of Cambridge Studies in Advanced Mathematics (1989)
\bibitem[MM]{MM} S. MacLane, I. Moerdijk - {\em Sheaves in Geometry and Logic: a first introduction to topos theory}. Springer Science $\&$ Business Media, 1992.
%\bibitem[HuKne]{HuKne} R. Huber, M. Knebusch, {\em On valuation spectra}, Contemporary Mathematics vol. 155 (1994).
%\bibitem[AM]{AM} M. F. Atiyah, I. G. MacDonald, {\em Introduction to Commutative Algebra}.
\bibitem[H1]{H1} R. Huber, {\em Continuous valuations}.
\bibitem[H2]{H2} R. Huber, {\em A generalization of formal schemes and rigid-analytic varieties}.
\bibitem[H3]{H3} R. Huber, {\em Bewertungsspektrum und rigid Geometrie}, Regensburger Mathematische Schriften 23, 1993.
%\bibitem[BV]{BuVe} K. Buzzard, A. Verberkmoes, {\em Stably uniform affinoids are sheafy}.
%\bibitem[SP]{Stacks} A. J. de Jong, {\em Stacks Project}.
%\bibitem[vdP]{vdP}  M. van der Put - {\em Cohomology on affinoid spaces.} Compositio Math. 45 (1982), 165 - 198.
%\bibitem[vdPS]{vdPS}  M. van der Put, P. Schneider - {\em Points of rigid analytic varieties.} J. reine angew. Math. 434 (1993), 127 - 157.
%\bibitem[Zar2]{Zar2} O. Zariski - {\em The reduction of the singularities of an algebraic surface}. Ann. Of Math. (2) 40 (3): 639-689 (1939).
%\bibitem[Zar3]{Zar3} O. Zariski {\em Reduction of the singularities of algebraic three dimensional varieties}. Ann. Of Math. (2) 45 (3) (1944).
%\bibitem[Wed]{Wed} T. Wedhorn, {\em Adic Spaces}, unpublished lecture notes.
%\bibitem[Bou]{Bou} N. Bourbaki, Alg\'ebre commutative, chap. I - VII, Springer (1989); chap. VIII- IX, Masson (1983); chap. X, Masson (1998).
%\bibitem[Sch]{IHES} P. Scholze, {\em Perfectoid spaces}.
\end{thebibliography}

\normalsize
\section{Lecture 15: Points and lft maps} 



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\subsection{A noetherian property}

Let $A$ be an affinoid algebraic over a non-archimedean field $k$ and $X = {\rm{Spa}}(A,A^0)$.  For any $x \in X$, the stalk
$\cO_{X,x}$ is the limit of the directed system $\{A_i\}$ of affinoid algebraic corresponding
to rational domains ${\rm{Spa}}(A_i, A_i^0) \subset X$ around $x$.  In particular, if $x$ is a classical point (i.e., corresponds
to a maximal ideal of $A$) then this limit likewise compute the stalk at $x$ on ${\rm{Sp}}(A)$, so by classical
rigid geometry \cite[7.3.2/7]{BGR} we know that $\cO_{X,x}$ is noetherian for such $x$ (with completion that naturally coincides
with the completion of $A$ at the maximal ideal associated to $x$).   But for more general $x$ we don't have the link to
rigid geometry and hence it is less clear if the stalk is noetherian.  Oddly, this result does not seem to be addressed in the literature,
so now we give Temkin's elegant proof of:

\begin{prop} For any $x \in X = {\rm{Spa}}(A,A^0)$, the local ring $\cO_{X,x}$ is Noetherian.
\end{prop}

\begin{proof}
The local ring at a point $x$ in an affinoid adic space $\Spa(A,A^+)$ for a classical affinoid algebra $(A,A^+)$ is the direct limit of a directed system $(A_i)$ for rational domains $\Spa(A_i,A^+)$ that contain $x$, and the transition maps $A_i \to A_j$ are flat by classical rigid-analytic geometry
\cite[7.3.2/6]{BGR}. 
Let $x_i \in \Spec(A_i)$ be the prime coming from the maximal ideal of the local limit $\cO_{X,x}$, so the directed system of algebraic localizations
$$B_i = (A_i)_{x_i}$$
has the same (local) limit.  Let $\fm_i$ denote the maximal ideal of $B_i$.

The advantage of considering the directed system ${B_i}$ is that since these are local noetherian
rings with flat local transition maps, by \cite[0$_{\rm{III}}$, 10.3.1.3]{EGA} the limit is noetherian if for sufficiently large $i$ we have 
$$\fm_i B_j = \fm_j$$
for all $j \ge i$. (This is the mechanism by which the noetherian property of henselizations of local noetherian rings is proved in \cite{EGA}, for example.)

Observe that $\dim(B_i) \leq \dim(A)$ for all $i$ since for a maximal ideal $\mathfrak{n}$ of $A_i$ containing $x_i$ and the point $\xi \in
{\rm{Sp}}(A_i) \subset {\rm{Sp}}(A) = X$ corresponding to $\mathfrak{n}$ we have
$\dim (B_i) \le \dim (A_i)_{\mathfrak{n}} = \dim \cO_{X,\xi} \leq \dim(A)$ (using that the local noetherian rings
$(A_i)_{\mathfrak{n}}$ and $\cO_{X,\xi}$ have the same completion
and hence the same dimension).  The dimension formula for flat local maps of local noetherian rings gives that 
$$\dim(B_j) = \dim(B_i) + \dim(B_j/\fm_i B_j) \geq \dim(B_i)$$
for all $j \geq i$.  Hence, the dimensions of the $B_i$'s are monotone in $i$ and bounded above, so 
for some large enough $i_0$ we have $\dim(B_i) = \dim(B_{i_0})$ for all $i \geq i_0$.  

Returning to the dimension formula,
the local noetherian fiber algebras $B_i/\fm_{i_0} B_j$ are 0-dimensional for all $i \geq i_0$.  
Thus, these fiber algebras have nilpotent maximal ideal, so to conclude that $\fm_i = \fm_{i_0} B_i$ for all $i \geq i_0$ it suffices 
to show that all fiber algebras of $\Spec(A_i) \to \Spec(A_{i_0})$ (e.g., the closed fiber, whose algebra admits $B_i/\fm_{i_0} B_i$ as an algebraic
localization) are reduced.  

More generally, for any map $A \to A'$ of classical affinoid algebras corresponding to an affinoid subdomain 
$\Sp(A') \hookrightarrow \Sp(A)$, and any $y'$  in $\Spec(A')$ and its image $y$ in $\Spec(A)$ we claim that for the local map
$$
(*)  \ \ \ A_x \longrightarrow A'_{y'}$$
all fibers algebras are geometrically regular, i.e., regular and remain so after any finite extension of the base field of the fiber algebra.  In other words, we claim that $(*)$ is a regular map of Noetherian rings.
Pick a maximal ideal $\fm'$ of $A'$ containing $y'$, so it contracts to a maximal ideal $\fm$ of $A$ containing $y$.  The map $(*)$ is a localization (on source and target) of the map
$$A_{\fm}\longrightarrow A'_{\fm'},$$
so it suffices to show that the latter flat local map is regular.  We know that the induced map on completions is an isomorphism by the classical theory, and to check the regularity property for a map of Noetherian rings it suffices to do after composing with a faithfully flat extension of the target
(since a noetherian ring is regular if it admits a faithfully flat regular extension ring).  Hence, it suffices to show the map from $A_{\fm}$ to its completion 
is a regular map. But this latter regularity is a consequence (and a key part in the proof) of the excellence of affinoid algebras,
an important result due to Kiehl.
\end{proof}









%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\subsection{Points as morphisms}

Given a morphism of adic spaces $f:X\to Y$, and a point $y\in Y$, it is natural to want to view
the fiber $f^{-1}(y)$ as the underlying topological space of an ``adic space fiber''.  However, matters
are not quite as for schemes because in the context of adic spaces a field (equipped with extra structure) generally does
not give rise to a one-point space!  So the correct expectation is
that $f^{-1}(y)$ should coincide with the fiber over the unique closed point in a suitably-defined
``adic space fiber'' over $\Spa(\kappa(y),\kappa(y)^+)$ where $\kappa(y)$ is the residue field of $\cO_{Y,y}$ and 
$\kappa(y)^+$ is its valuation ring (the image of $\cO_{Y,y}^+$ in $\kappa(y)$); of course, we could also replace $\kappa(y)$ and $\kappa(y)^+$ with
their completions for this purpose.  Strictly speaking, to make sense of $\Spa(\kappa(y), \kappa(y)^+)$ we have to know
that $(\kappa(y), \kappa(y)^+)$ is a Huber pair, and we saw in 14.6 that this holds when $y$ is an analytic point.

Since we have not discussed fiber products of adic spaces (a topic which has some subtleties, since in contrast with schemes
it will not always exist), the preceding discussion merely serves as geometric motivaton for the algebraic considerations that follow.
In effect, just as a point of an affine scheme can be interpreted in terms of maps from the given ring to a field, we seek a similarly intepretation
for points of affinoid adic spaces.

Let $(A,A^+)$ be a Huber pair, $X:=\Spa(A,A^+)$ and $\Spa(A',{A'}^+)$ a rational domain of $X$ containing a point $x\in X$. Let $\fp_x$ be the support of $x$ in $A$ and set $\kappa(x):=\Frac(A/\fp_x)$ as a valued field using $v_x$. The following diagram is commutative:

\begin{displaymath}
\xymatrix{\kappa(x)\ar@{=}[d]\ar@{^(->}[r] & \kappa'(x)\ar@{=}[d]\\
		\Frac(A/\fp_x)\ar[d]_{v_x} & \Frac(A'/\fp_x')\ar[dl]^{v_x'}\\
		\Gamma_x & }
\end{displaymath}

Recall that $\kappa(x)$ and $\kappa'(x)$ have the same completion, as discussed in Lecture 14. In any global situation $\kappa(x)$ will mean 
$\cO_x/\fm_x$ or $(\cO_x/\fm_x)^{\wedge}$ (the context should make the intent clear). The
completion convention is, given the last observation, the safest to work with as it is computed using any affinoid open subspace around $x$. 

Let us now assume $x$ is an analytic point of $X$, so every generization of $x$ is analytic (as analyticity is an open condition)
and $(\kappa(x), \kappa(x)^+)$ is a Huber pair.  By Proposition 8.3.10, all generizations of $x$ in $X$ are {\em vertical} generizations.

Analyticity of $x$ ensures that $\kappa(x)$ with its $v_x$-topology is topologically a rank-1 valued field, where the rank-1 valuation ring 
is the subring $\kappa(x)^0$ of power-bounded elements (generally bigger than $\kappa(x)^+$).
Recall that there is a unique rank-1 vertical generization $\eta_x \in \Spv(\kappa(x))$ of $x$ that lies in the fiber through $x$
for $X\to\Spec(A)$, and that in general (by \S4.2 in Lecture 4 and \cite[10.1]{Mat}) the set
$$\{\text{generizations of}\ x \text{ in}\ \Spv(\kappa(x))\}$$
is in bijection with the set of valuation subrings $R$ of $\kappa(x)$ containing $\kappa(x)^+$ 
as well as with the set $\Spec(\kappa(x)^+)$ via the correspondence assigning to any prime ideal $\mathfrak{p}$ of $\kappa(x)^+$
the localization $R = \kappa(x)^+_{\mathfrak{p}}$. 

The set of all generizations of $x$ in $X$ (not just in $\Spv(A)$!) is therefore $$\Spa(\kappa(x),\kappa(x)^+)=\{\kappa(x)^+\subset R\subset\kappa(x)^0\}$$
under which $x$ is the unique closed point and $\eta_x$ is the unique generic point (corresponding to $\kappa(x)^0$); this does {\em not}
contain the generic point of $\Spv(\kappa(x))$ (i.e., the trivial valuation) since the latter is not continuous on $A$ and so does not lie in $X$ inside
$\Spv(A)$.
We are now going to focus on $\Spa(\kappa(x),\kappa(x)^+)$; as the topological space of all generizations of $x$ in $X$
(with the subspace topology from $X$) it is an analogue of $\Spec(\cO_{X,x})$ in the theory of schemes. 

Since for analytic $x \in X$ as above the valuation subring $\kappa(x)^+$ is an open subring of the rank-1 valuation ring
$\kappa(x)^0$, we are motivated to make:

\begin{defin} \label{affinoid fld} An {\em affinoid field} $(k,k^+)$ is a pair consisting in a complete nonarchimedean field $k$ and an 
open valuation subring $k^+$ of $k^0$ (so ${\rm{Frac}}(k^+) = k$ due to openness of $k^+$ in $k$). 
\end{defin}

\begin{rem} Any affinoid field $(k,k^+)$ is a Huber pair.  Indeed, since $k^0$ is an intermediate valuation ring between
$k^+$ and $k = {\rm{Frac}}(k^+)$, necessarily $k^0$ arises from a valuation on the residue field of $k^+$ by
\cite[10.1]{Mat} and so the maximal ideal $k^{00}$ of $k^0$ is contained in that of $k^+$.  Thus, if $\varpi \in k^0$ is a pseudo-uniformizer
and $x \in k^{\times}$ then $\varpi^n/x \in k^{00} \subset k^+$ for all large $n$
(largeness depending on $x$), so $v(\varpi)^n \le v(x)$ where $v$ denotes the valuation for $k^+$.  It follows that
$v(\varpi^n k^+) \le v(x)$ for all large $n$, so the open ideals $\varpi^n k^+$ in $k^+$ are a base for the $v$-topology on $k$,
thereby verifying the Huber property.
\end{rem}

\begin{prop} \label{analytic bijection} Let $(A,A^+)$ be any Huber pair, and consider the set
$$\Sigma_{A}:=\{(A,A^+)\xrightarrow{\varphi}(k,k^+)\mid\varphi(A)\subset k\ \text{\rm{generates a dense subfield}}\}$$
where $(k,k^+)$ is some affinoid field, or equivalently the set of continuous valuations $v$ on $A$ for which $v(A^+) \le 1$ and 
the associated completion of ${\rm{Frac}}(A/\mathfrak{p}_v)$ has the topology of a rank-$1$ valued field.

There is a natural bijection
$$\Sigma_A\simeq|\Spa(A,A^+)_{\rm an}|$$
assigning to any $\varphi \in \Sigma_A$ the valuation $v_{\varphi}$ that is the image under $\Spa(\varphi)$ of the unique closed point in
$\Spa(k, k^+)$.
\end{prop}

\begin{proof} This is immediate from the preceding discussion since the valuation on $k$ associated to $k^+$ is the unique
closed point in $\Spa(k, k^+)$.
\end{proof}

As a corollary we can immediately globalize by working with affinoid open subspaces:

\begin{cor} For an adic space, the set $|X_{\rm{an}}|$ of analytic points is in natural bijection with the set of morphisms
$$\{f:\Spa(k,k^+)\to X\mid \kappa(x)\subset k\ \text{\rm{is dense}}\}$$
from the adic spectra of affinoid fields to $X$ by assigning to any such $f$ the image of the unique closed
point of $\Spa(k,k^+)$.
\end{cor}

This corollary is most useful when {\em all} points of $X$ are analytic. Since any map of adic spaces
carries analytic points to analytic points, if $X$ is equipped with a morphism to an adic space $S$ whose points are all
analytic then the same holds for $X$.  For instance, this happens when $S = \Spa(K, K^+)$ for an affinoid field
$(K, K^+)$, the most important example being when $K^+ = K^0$.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\subsection{Locally of finite type maps}

In this section we discuss what a locally finite type morphism of adic spaces should be.  
One wrinkle is that classical rigid geometry over a complete nonarchimedean field $k$
does not provide much guidance, since (in hindsight) {\em all} morphisms in classical
rigid geometry are locally of finite type when upgraded to morphisms of adic spaces (since the adic spaces themselves
will be locally of finite type over $\Spa(k, k^0)$).

Let $(A,A^+)$ and $(B,B^+)$ be {\em complete} Huber pairs, with $X:=\Spa(B,B^+)$ and $S:=\Spa(A,A^+)$, and let $f:X\to S$ be a morphism. 
One context in which we certainly want to say that $f$ is ``finite type'' is if $f$ can be factored as a ``closed immersion'' into some relative unit
polydisc over $S$.  On the level of Huber pairs, this amounts to having a commutative diagram
\begin{displaymath}
\xymatrix{(A\langle y_1,\dots, y_n\rangle, A\langle y_1,\dots, y_n\rangle^+) \ar[r] & (B, B^+) \\
	(A,A^+)\ar[ur] \ar[u] & }
\end{displaymath}
where $A\langle y_1,\dots, y_n\rangle \rightarrow B$ is a topological quotient map (i.e., $B$ is topologically the quotient of
$A\langle y_1, \dots, y_n \rangle$ by a closed ideal) 
and the open image $B'$ of $A\langle y_1, \dots, y_n \rangle^+$ in $B$ is a ring of definition whose integral closure is $B^+$. 
(Here it is no different to work with the open subring $A^+\langle y_1, \dots, y_n \rangle$ whose integral closure in
$A \langle y_1, \dots, y_n \rangle$ in $A\langle y_1, \dots, y_n \rangle^+$ by definition.)

We should allow ``varying polyradii'' in these relative polydiscs, much as in classical rigid geometry we consider
polydiscs given by conditions $\{|y_i| \le r_i\}$ for $r_i \in \surd{|k^{\times}|}$ (not requiring $r_i \in |k^{\times}|$). 
So for nonempty finite subsets $T_1,\dots, T_n\subset A$ such that $T_iA\subset A$ is open we should also allow ourselves to use 
the topological $A$-algebra
$$A\langle X\rangle_T:=\{\sum a_{\nu} X^{\nu} \in A[\![X]\!] 
\mid \text{\rm{for every }} U, a_{\nu} \in T^{\nu} U \text{\rm{ for all but finitely many }} \nu\}$$
where $U$ varies through additive open subgroups of $A$, having as 
a base of opens $$U\langle X\rangle_T:=\{\sum a_{\nu}X^{\nu}\in A\langle X\rangle_T\mid a_{\nu}\in T^{\nu}U 
\text{\rm{ for all }} \nu\}$$
and ring of definition $A^+\langle X \rangle_T$.
We recall that $A\langle X\rangle_T$ is Huber, complete when $A$ is complete,
and is initial among Huber rings over $A$ equipped with an ordered $n$-tuple
of elements $x_1,\dots, x_n$ such that $t_ix_i$ is power-bounded for all $t_i\in T_i$ and for all $1 \le i \le n$.

Finally, given a Huber pair $(A,A^+)$, we define $A\langle X\rangle_T^+$ to be the integral closure of the 
ring of definition $A^+\langle X\rangle_T\subset A\langle X\rangle_T$.

\begin{defin} \label{tft} Given a map $(A,A^+)\to (B,B^+)$ of complete Huber pairs, we say it is 
{\em  of finite type} if there exists $T_1, \dots, T_n$ as above and a map of Huber pairs
$$(A\langle X\rangle_T, A\langle X\rangle_T^+) \to (B, B^+)$$
such that the map on Huber rings is an open surjection and the map $A\langle X \rangle_T^+ \rightarrow B^+$ is integral;
equivalently, $B = A\langle X \rangle_T/J$ for some closed ideal $J$ and $B^+$ is the integral closure in $B$
of the open image of $A^+\langle X\rangle_T$.
\end{defin}

\begin{eg} \label{eg tft} For a complete Huber pair $(A, A^+)$ and an element $s \in A$ and a non-empty finite subset $T = \{t_1,\dots, t_n\}
\subset A$ such that
$T \cdot A \subset A$ is open, consider the rational subdomain
$$A\langle T/s\rangle=A\langle X_1, \dots, X_n\rangle_T/\overline{(sX_i-t_i)}$$
with $A\langle T/s\rangle^+$ as usual. Then $(A\langle T/s\rangle, A\langle T/s\rangle^+)$ is of  finite type over $(A,A^+)$. 
\end{eg}

It is important to note that when the base ring is Tate (e.g., a non-archimedean field), the role of the polyradii can be suppressed:

\begin{prop}\label{tateft} If $A$ is Tate then for $(B,B^+)$  of finite type over $(A,A^+)$ we can always choose $T_i=\{1\}$ for all $i$;
i.e., for any $T_1, \dots, T_n$ there exists a continuous open 
$A$-algebra surjection $A\langle Y \rangle \twoheadrightarrow A\langle X\rangle_T$
such that $A^+\langle Y \rangle \rightarrow A^+\langle X\rangle_T$ is integral.
\end{prop}

\begin{proof}  The idea is to identify $\Spa(A\langle X\rangle_T, A\langle X\rangle_T^+)$ with a rational domain in some 
$\Spa(A\langle Y\rangle, A\langle Y\rangle^+)$, as then we can apply Example \ref{eg tft} to conclude.

First we shall make a ``change of coordinates'' to bound the $T_i$'s relative to a ring of definition, as follows.
Let $A_0 \subset A$ be a ring of definition, and 
pick a topologically nilpotent unit $\varpi\in A$.  Let $m \ge 0$ be an integer large enough so that
$\varpi^mt_i\in A_0\subset A$ for all $t_i \in T_i$ ($1 \le i \le n$). 
By replacing $X_i$ with $\varpi^{-m}X_i$ and replacing $T_i$ with $\varpi^mT_i$ for all $i$, 
we may thereby assume $t_i\in A_0$ for all $t_i\in T_i$ and for all $i$.

Recall that $A\langle X\rangle_T$ is universal among Huber rings over $A$ equipped
with an ordered $n$-tuple of elements $x_1, \dots, x_n$ such that
$t_ix_i$ is power-bounded for all $t_i\in T_i$ and all $i$. We have rigged that all elements of each $T_i$
are power-bounded, and the only open ideal in $A$ is the unit ideal (by the Tate hypothesis!), so $T_i A = A$ for all $i$. 
Letting $T_i X_i$ denote the finite subset of elements $t_i X_i \in A \langle X \rangle$,
we get a finite subset 
$$\Sigma:=T_1\cup\dots\cup T_n\cup T_1X_1\cup\dots\cup T_nX_n \subset A\langle X\rangle^0$$
that generates an open ideal of $A\langle X\rangle$ (even the unit ideal).  Therefore it makes sense to form the completed rational localization
$$(A\langle X\rangle)\langle \Sigma/1 \rangle$$
that is a Huber ring and uniquely isomorphic to $A\langle X\rangle_T$
as a topological $A\langle X \rangle$-algebra via their universal properties as
topological $A$-algebras. It is straightforward to chase the ``integral structures'' to check that this achieves
the original idea with a rational domain in a relative unit polydisc over $\Spa(A, A^+)$.
\end{proof}

As an application of the preceding proposition, we get a first result towards characterizing adic spaces associated
to affinoid algebras over a non-archimedean field $k$ 
among general adic spaces over $\Spa(k, k^0)$:

\begin{cor} \label{tft over aff field} Let $k$ be a nonarchimedean field, $A$  a $k$-affinoid algebra, and $(A,A^+)$ is a Huber pair, 
so $A^+\subset A^0$ is an open subring. The map $(k,k^0)\to (A,A^+)$ is  of finite type if and only if $A^+=A^0$.
\end{cor}

\begin{proof} Since there exists an open surjection 
$$k\langle X\rangle \twoheadrightarrow k\langle X\rangle/J\simeq A$$
and we have seen that the associated map $k\langle X\rangle^0\to A^0$ is integral, 
the ``if'' direction is verified.  Conversely, assume $(k,k^0)\to (A,A^+)$ is  of finite type, so by
Proposition \ref{tateft} over the base $(k, k^0)$ there exists an open map
$$k\langle X\rangle \twoheadrightarrow A$$
which induces an integral map
$$k\langle X\rangle^0 \to A^+.$$
But composing it with the inclusion $A^+\subset A^0$ is known to give an integral map
(as always for integral maps, such as surjections, between $k$-affinoid algebras), so $A^+ \rightarrow A^0$
is integral.  Since $A^+$ integrally closed in $A$, we get $A^+=A^0$ as desired.
\end{proof}



\begin{defin} A morphism of adic spaces $f : X \to Y$ 
is {\em locally of  finite type} if, for any $x\in X$, there exists an affinoid open subspaces $U$ around $x$ and 
$V$ around $f(x)$ such that $f(U)\subseteq V$ and the induced homomorphism of 
complete Huber pairs $(\cO_Y(V), \cO^+_Y(V))\to (\cO_X(U),\cO^+_X(U))$ 
is  of finite type as in Definition \ref{tft}.
\end{defin}

By Example \ref{eg tft}, the open immersion of a rational domain into an affinoid adic space is
locally of  finite type.  Thus, in general {\em every} open immersion of adic spaces
is locally of  finite type since the rational domains give a base of open subspaces of an
affinoid adic space.  

It is clear that in order for the notion ``locally of  finite type'' to be useful, in the affinoid setting it has to coincide
with ``of finite type''.  The analogous fact for schemes is quite elementary, but for adic spaces
the proof is much harder (involving tricky approximation arguments) and is an important theorem of Huber: 

\begin{thm} Assume that $(A,A^+)$ and $(B,B^+)$ are complete Huber pairs such that each of $A$ and $B$ is strongly Noetherian, or each has Noetherian ring of definition. Then a map $(A,A^+)\to (B,B^+)$ which is locally of finite type is in fact of finite type.
\end{thm}

The only published proof appears to be \cite[Thm.\,3.8.15]{H3}, and unfortunately \cite{H3} is difficult to obtain 
(e.g., the only copy in the United States is in the University of Michigan math library).  
In a later version of these notes I will include a proof (in English).



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%
%\subsection{Appendix A: rigid varieties and adic spaces}
%
%To construct a functor $\mathfrak{R}_k\rightarrow\mathfrak{A}_k$ we find convenient to stick to the more abstract setting of ringed sites, and thus we may view a morphism of adic spaces $X\rightarrow Y$ as inducing a morphism of ringed sites $(|X|,\cO_X)\rightarrow(|Y|,\cO_Y)$.
%\paragraph{Construction of the functor $(-)^{\rm ad}:\mathfrak{R}_k\rightarrow\mathfrak{A}_k$} \ \\
%
%Let us notice that if $X=\Sp(k)$ and $Y=\Spa(k,k^{0})$, then we have an equivalence of ringed sites
%$$(|X|,\cO_X)=(|Y|,\cO_Y)$$
%We call this site $S$. We now discuss some of the details of Huber's proof of the following
%\begin{prop} \label{eta} Let $X\in\textup{ob}(\mathfrak{R}_k)$. There exists an adic space $X^{\rm ad}\in\textup{ob}(\mathfrak{A}_k)$ and a morphism
%\begin{displaymath}
%\xymatrix{(|X^{\rm ad}|,\cO_{X^{\rm ad}})\ar[r]^{\eta_X} & (|X|,\cO_X)}
%\end{displaymath}
%of ringed sites over $S$, such that $X^{\rm ad}$ is locally of finite type over $\Spa(k,k^{0})$, $\eta$ is locally coherent, and the following universal property holds true. If $Z$ is an adic space locally of finite type over $\Spa(k,k^{0})$ and 
%\begin{displaymath}
%\xymatrix{(|Z|,\cO_Z)\ar[r]^{\vartheta} & (|X|,\cO_X)}
%\end{displaymath}
%is a locally coherent morphism of ringed sites over $S$, then there exists a unique morphism
%\begin{displaymath}
%\xymatrix{Z\ar[r]^f & X^{\rm ad}}
%\end{displaymath}
%in the category $\mathfrak{A}_k$, making the following diagram commute:
%\begin{displaymath}
%\xymatrix{(|Z|,\cO_{Z})\ar[r]^{f^{\rm s}}\ar[rd]_{\vartheta} & (|X^{\rm ad}|,\cO_{X^{\rm ad}})\ar[d]^{\eta_X}\\
%							& (|X|,\cO_X)}
%\end{displaymath}
%\end{prop}
%Before discussing the proof, let us state and prove one more result which eventually assigns the functor $\mathfrak{R}_k\rightarrow\mathfrak{A}_k$.
%\begin{prop} Let $f:X\rightarrow Y$ be a morphism in the category $\mathfrak{R}_k$. Then the induced morphism of ringed sites over $S$
%\begin{displaymath}
%\xymatrix{(|X|,\cO_X)\ar[r]^{f^{\rm s}} & (|Y|,\cO_Y)}
%\end{displaymath}
%is locally coherent.
%\end{prop}
%\begin{proof}
%For every quasi-compact open subset $U\subset X$, the restriction $f\vert_U$ is quasi-compact as a morphism in the category of adic spaces over $\Spa(k,k^{0})$, meaning that the morphism $|X|\rightarrow|Y|$ on the level of $S$-sites is locally coherent. It obviously respect the ringed structures $\cO_X$ and $\cO_Y$ on $|X|$ and $|Y|$.
%\end{proof}
%Indeed, by Proposition \ref{eta}, we have the natural morphisms $\eta_X$ and $\eta_Y$. If we choose $Z=X^{\rm ad}$ and $\vartheta=f^{\rm s}\circ\eta_X$, by the universal property we have as in the following diagram
%\begin{displaymath}
%\xymatrix{(|X^{\rm ad}|,\cO_{X^{\rm ad}})\ar[d]_{\boxed{\exists ! h}}\ar[r]^{\eta_X} & (|X|,\cO_X)\ar[d]^{f^{\rm s}}\\
%		(|Y^{\rm ad}|,\cO_{Y^{\rm ad}})\ar[r]_{\eta_Y} & (|Y|,\cO_Y)}
%\end{displaymath}
%If we set $f^{\rm ad}:=h$, by universality of $h$ this indeed says $(-)^{\rm ad}$ is a functor $\mathfrak{R}_k\rightarrow\mathfrak{A}_k$.\\ \ \\
%Now we sketch the proof of Proposition \ref{eta}.
%\begin{proof}[Sketch of proof of Proposition \ref{eta}]
%We can assume $X$ is affinoid, ie. $X=\Sp(A)$ for some Tate tft $k$-algebra $A$. Let us pick $x\in A$. It corresponds to a maximal ideal $\fm_x$ of $A$. Let $k':=A/\fm$. $k'/k$ is a finite extension, and therefore we can uniquely extend the valuation on $k$ to $k'$. The composite
%\begin{displaymath}
%\xymatrix{A\ar[r]\ar@/_1.3pc/[rr] & A/\fm =k'\ar[r] & \bR^+_{>0}\cup\{0\}}
%\end{displaymath}
%assigns a valuation on $A$. Since $A$ is a Tate $k$-algebra, such a valuation will be $\leq 1$ on $A^{0}$, and therefore defines a point of $\Spa(A,A^{0})$. We set $$X^{\rm ad}:=\Spa(A,A^{0}).$$
%This assigns a natural one-to-one correspondence between rational subsets of $X$ and rational subsets of $X^{\rm ad}$.
%Via this correspondence, given $U$ a rational subset of $X$ and $V$ the corresponding rational subset of $X^{\rm ad}$, we have isomorphisms of $A$-algebras
%\begin{displaymath}
%\xymatrix{\cO_X(U)\ar[r]^{\varphi_U} & \cO_{X^{\rm ad}}(V)}
%\end{displaymath}
%The rational subsets form a basis for the $G$-topology on $X$, as well as for the topology on $X^{\rm ad}$. Therefore the correspondence between rational subsets of $X$ and $X^{\rm ad}$ respectively, actually extends uniquely to a morphism of sites
%\begin{displaymath}
%\xymatrix{|X|\ar[r]^{\psi} & |X^{\rm ad}|}
%\end{displaymath}
%so that we can produce also sheaf isomorphisms
%\begin{displaymath}
%\xymatrix{\cO_X\ar[r]^{\varphi\ \ \ \ \ }_{\sim\ \ \ \ \ } & \psi_*(\cO_{X^{\rm ad}})}
%\end{displaymath}
%We define
%$$\eta_X:=(\psi,\varphi).$$
%This is indeed a morphism of ringed sites $(|X^{\rm ad}|,\cO_{X^{\rm ad}})\rightarrow (|X|,\cO_X)$ over $S$. Finally, one just checks the universal property, given a locally coherent morphism $\vartheta:(|Z|,\cO_Z)\rightarrow (|X|,\cO_X)$ in the category of ringed sites. Existence of $f^{\rm s}$ is given aready in \cite{SGA4}, on the level of ringed sites (here we use $\vartheta$ is locally coherent). Since, actually, $Z$ is an adic space, $f^{\rm s}$ actually descends to a morphism of ringed {\em spaces} $f:(Z,\cO_Z)\rightarrow(X^{\rm ad},\cO_{X^{\rm ad}})$, and we are left to show that $f$ is a morphism in $\mathfrak{A}_K$. The argument eventually makes use of the finiteness assumption on $Z$ as an adic space over $\Spa(k,k^{0})$ to use the fact that $k$-algebra homomorphisms between Tate rings of tft over $k$ are continuous.
%\end{proof}
%
%\paragraph{$(-)^{\rm ad}$ embeds $\mathfrak{R}_k$ as a full subcategory of $\mathfrak{A}_k$}\ \\
%
%We have indeed already proved this, if we consider Proposition \ref{eta} carefully. The key step in defining $X^{\rm ad}$ as an adic space is in the assignement
%$$\Sp(A)\mapsto\Spa(A,A^{0})$$
%which was suggested by simply considering the points of $\Sp(A)$, and observing that they give rise to a point in $\Spa(A,A^{0})$. If, again, we pick a point $x\in \Sp(A)$, and consider the corresponding maximal ideal $\fm_x\subset A$, we are naturally given a finite field extension $A/\fm_x$ of $k$. The crucial step is that there exists a unique valuation on $A/\fm_x$ extending the valuation on $k$! In other words, due to this, $X^{\rm ad}$ is uniquely determined by $X$ as an adic space! Functoriality of this assignement is given already on the level of ringed sites by \cite{EGA4}.
%
%\paragraph{An important consequence.}\ \\
%There is a key step we have not enough extensively discussed in the sketch of proof. Given a rigid-analytic variety $X$ over $k$, according to the Proposition $X^{\rm ad}$ is of finite type over $\Spa(k,k^{0})$. This is important, as it allows us to see that $X$ is an affinoid rigid-analytic variety over $k$ if and only if $X^{\rm ad}$ is an affinoid adic space over $\Spa(k,k^{0})$.\\
%In fact, let $X\in\textup{ob}(\mathfrak{R}_k)$ be such that $X^{\rm ad}$ is affinoid. Therefore, the morphism of affinoid rings
%$$(k,k^{0})\longrightarrow(\cO_{X^{\rm ad}}(X^{\rm ad}),\cO_{X^{\rm ad}}^+(X^{\rm ad}))$$
%is of tft. The adic space $X^{\rm ad}$ is isomorphic over $\Spa(k,k^{0})$ to
%$$\Spa(\cO_{X^{\rm ad}}(X^{\rm ad}),\cO_{X^{\rm ad}}(X^{\rm ad})^{\circ})=\Sp(\cO_{X^{\rm ad}}(X^{\rm ad}))^{\rm ad}.$$
%Call $A:=\cO_{X^{\rm ad}}(X^{\rm ad})$. Since $(-)^{\rm ad}$ is fully faithful, we can say $\Sp(A)$ is isomorphic to $X$ as a rigid-analytic variety over $k$ only if we have $A^+=A^{0}$. By this finiteness condition, this is indeed the case, by Corollary \cite{tft over aff field}
%



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%\subsection{Appendix B: formal schemes and adic spaces}
%
%
%We wish, now, to construct a functor from the category of formal schemes to the category of adic spaces. We prove the key results, and sketch some of the consequences of the main Theorem of this section, which the reader can find in \cite{H2}.
%
%\begin{defin} Let $\fX$ be a formal scheme that is adic in the sense that the topology on the coordinate ring of every affine open
%subspace is defined by a finitely generated ideal.  We say it is {\em sheafy} is for any affine 
%open subspace  $U=\Spf(A)\subset\fX$, the adic ring $A$ is such that $\Spa(A,A)$ is sheafy.
%\end{defin}
%
%Let $Y$ be an adic space. We can obviously consider $\cO_Y^+$ as a sheaf of topological rings by endowing $\cO_Y^+(U)$ with the subspace topology of $\cO_Y(U)$, for any $U\subset Y$ open. Then $(Y,\cO_Y^+)$ is still a locally and topologically ringed space. Every morphism of adic spaces $f:X\rightarrow Y$ induces a morphism $f^+:(X,\cO_X^+)\to(Y,\cO_Y^+)$ of locally and topologically ringed spaces. 
%
%\begin{prop} Let $\fX=\Spf(A)$ be a sheafy affine adic formal scheme. Then, for any adic space $Z$, the map
%$$\Hom((Z,\cO_Z^+),(\fX,\cO_{\fX}))\longrightarrow \Hom_{\rm cont}(A,\Gamma(Z,\cO_Z^+))$$
%sending a morphism $(f,f^{\sharp})$ of locally and topologically ringed spaces to a continuous homomorphism $f_{\fX}$ of adic rings, is a bijection.
%\end{prop}
%
%This gives some hints on how to define a functor from some category of adic formal schemes to the category of adic spaces.
%\begin{thm} Let $\fX$ be a sheafy adic formal scheme. There exists an adic space $\fX^{\rm ad}$ and a morphism of locally and topologically ringed spaces
%\begin{displaymath}
%\xymatrix{(\fX^{\rm ad},\cO_{\fX^{\rm ad}}^+)\ar[r]^{\alpha} & (\fX,\cO_{\fX})}
%\end{displaymath}
%satisfying the following universal property. For every adic space $Z$ and for every morphism $f:(Z,\cO_Z^+)\to (\fX,\cO_{\fX})$ of locally and topologically ringed spaces, there exists a unique morphism $f^{\rm ad}:Z\to\fX^{\rm ad}$ making the following diagram commute
%\begin{displaymath}
%\xymatrix{(Z,\cO_Z^+)\ar[rd]_{f^{\rm ad}}\ar[r]^f & (\fX,\cO_{\fX})\\
%				& (\fX^{\rm ad}, \cO_{\fX^{\rm ad}})\ar[u]_{\boxed{\alpha}}}
%\end{displaymath}
%\end{thm}
%\begin{proof}
%Morphisms of adic spaces glue together, so it suffices to 
%prove this when $\fX=\Spf(A)$. Define $$\fX^{\rm ad}:=\Spa(A,A).$$
%The identity $A\to A$ induces a morphism of locally and topologically ringed spaces $\alpha:(\fX^{\rm ad},\cO_{\fX^{\rm ad}})\to (\fX,\cO_{\fX})$. For every adic space $Z$, we have natural bijections
%\begin{flalign*}
%\Hom_{\mathbf{Adic}}(Z,\fX^{\rm ad})=&\ \Hom_{\mathbf{Affinoid}}((A,A),(\Gamma(Z,\cO_Z),\Gamma(Z,\cO_Z^+)))\\
%		=&\ \Hom_{\mathbf{TopRng}}(A,\Gamma(Z,\cO_Z^+))\\
%		=&\ \Hom_{\mathbf{LTRSp}}((Z,\cO_Z^+),(\fX^{\rm ad},\cO_{\fX^{\rm ad}}^+)).
%\end{flalign*}
%where, respectively, $\mathbf{Adic}$ is the category of adic spaces, $\mathbf{Affinoid}$ is the category of affinoid rings, $\mathbf{TopRng}$ is the category of topological rings and $\mathbf{LTRSp}$ is the category of locally and topologically ringed spaces. 
%It is easy to check that $\alpha$ and $\fX^{\rm ad}$ satisfy the required universal property
%since for any morphism of adic spaces $f:Z\to\fX^{\rm ad}$, we apply the bijections above 
%to the composition $\alpha\circ f^+$ to obtain the desired $f^{\rm ad}$.
%\end{proof}
%
%As a consequence, $\fX\mapsto \fX^{\rm ad}$ assigns a functor from the category of sheafy adic formal schemes to the category of adic spaces. 
%In fact, let $f:\fX\to\fY$ be a morphism of sheafy  adic formal schemes. The universal property applied to the morphism
%\begin{displaymath}
%\xymatrix{(\fX^{\rm ad},\cO_{\fX^{\rm ad}}^+)\ar@/_1.0pc/[rr]\ar[r]^{\alpha} & (\fX,\cO_{\fX})\ar[r]^f & (\fY,\cO_{\fY})}
%\end{displaymath}
%yelds a unique morphism $f^{\rm ad}:\fX^{\rm ad}\to\fY^{\rm ad}$ of adic spaces, making the following diagram commute in the category of locally and topologically ringed spaces
%\begin{displaymath}
%\xymatrix{(\fX^{\rm ad},\cO_{\fX^{\rm ad}}^+)\ar[d]_{\alpha_{\fX}}\ar[r]^{f^{\rm ad +}} & (\fY^{\rm ad},\cO_{\fY^{\rm ad}}^+)\ar[d]^{\alpha_{\fY}}\\
%		(\fX,\cO_{\fX})\ar[r]_f & (\fY,\cO_{\fY})}
%\end{displaymath}
%Uniqueness of $f^{\rm ad}$ implies its functorialty, so $\fX\mapsto\fX^{\rm ad}$ is a functor.
%
%\paragraph{A special example}\ \\
%
%Just to fix ideas on how different are adic spaces from formal schemes, let us consider the following very basic example. We let $A$ be an adic ring with finitely generated ideal of definition $I$. We define
%$$X_{\rm triv}:=\{x\in\Spa(A,A)\mid v_x\ \textup{is trivial}\}.$$
%Then the map sending $v_x$ to its support yields a homeomorphism
%$$X_{\rm triv}\simeq\Spf(A).$$
%The inclusion $X_{\rm triv}\subset X$ is spectral, because, explicitly, the intersection of a rational subset $U\left(\frac{f_1,...,f_n}{g}\right)$ with $X_{\rm triv}$ is just $D^+(g)$!
%To sum up, we have the following
%\begin{thm} The functor $\fX\mapsto \fX^{\rm ad}$ from the category of locally noetherian formal schemes to the category of adic spaces is fully faithful.
%\end{thm}
%
%





%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\small

















\end{document}